UC Davis UC Davis Previously Published Works

Title

Skeeter Buster: A Stochastic, Spatially Explicit Modeling Tool for Studying Aedes aegypti Population Replacement and Population Suppression Strategies

Permalink https://escholarship.org/uc/item/7ss208dv

Journal PLOS Neglected Tropical Diseases, 3(9)

ISSN 1935-2727

Authors

Magori, Krisztian Legros, Mathieu Puente, Molly E <u>et al.</u>

Publication Date 2009

DOI

10.1371/journal.pntd.0000508

Peer reviewed

Skeeter Buster: A Stochastic, Spatially Explicit Modeling Tool for Studying *Aedes aegypti* Population Replacement and Population Suppression Strategies

Krisztian Magori^{1,9^a}, Mathieu Legros^{1,9}*, Molly E. Puente¹, Dana A. Focks², Thomas W. Scott³, Alun L. Lloyd⁴, Fred Gould¹

1 Department of Entomology, North Carolina State University, Raleigh, North Carolina, United States of America, 2 Infectious Disease Analysis, Gainesville, Florida, United States of America, 3 Department of Entomology, University of California, Davis, California, United States of America, 4 Department of Mathematics and Biomathematics Graduate Program, North Carolina State University, Raleigh, North Carolina, United States of America

Abstract

Background: Dengue is the most important mosquito-borne viral disease affecting humans. The only prevention measure currently available is the control of its vectors, primarily *Aedes aegypti*. Recent advances in genetic engineering have opened the possibility for a new range of control strategies based on genetically modified mosquitoes. Assessing the potential efficacy of genetic (and conventional) strategies requires the availability of modeling tools that accurately describe the dynamics and genetics of *Ae. aegypti* populations.

Methodology/Principal findings: We describe in this paper a new modeling tool of *Ae. aegypti* population dynamics and genetics named Skeeter Buster. This model operates at the scale of individual water-filled containers for immature stages and individual properties (houses) for adults. The biology of cohorts of mosquitoes is modeled based on the algorithms used in the non-spatial Container Inhabiting Mosquitoes Simulation Model (CIMSiM). Additional features incorporated into Skeeter Buster include stochasticity, spatial structure and detailed population genetics. We observe that the stochastic modeling of individual containers in Skeeter Buster is associated with a strongly reduced temporal variation in stage-specific population densities. We show that heterogeneity in container composition of individual properties has a major impact on spatial heterogeneity in population density between properties. We detail how adult dispersal reduces this spatial heterogeneity. Finally, we present the predicted genetic structure of the population by calculating F_{ST} values and isolation by distance patterns, and examine the effects of adult dispersal and container movement between properties.

Conclusions/Significance: We demonstrate that the incorporated stochasticity and level of spatial detail have major impacts on the simulated population dynamics, which could potentially impact predictions in terms of control measures. The capacity to describe population genetics confers the ability to model the outcome of genetic control methods. Skeeter Buster is therefore an important tool to model *Ae. aegypti* populations and the outcome of vector control measures.

Citation: Magori K, Legros M, Puente ME, Focks DA, Scott TW, et al. (2009) Skeeter Buster: A Stochastic, Spatially Explicit Modeling Tool for Studying *Aedes aegypti* Population Replacement and Population Suppression Strategies. PLoS Negl Trop Dis 3(9): e508. doi:10.1371/journal.pntd.0000508

Editor: Pattamaporn Kittayapong, Mahidol University, Thailand

Received March 6, 2009; Accepted July 27, 2009; Published September 1, 2009

Copyright: © 2009 Magori et al. This is an open-access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Funding: This work is funded by a grant to the Regents of the University of California from the Foundation for the National Institutes of Health (NIH) through the Grand Challenges in Global Health initiative and an NIH grant R01-AI54954-0IA2 to FG. The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing Interests: The authors have declared that no competing interests exist.

* E-mail: mlegros@ncsu.edu

¤ Current address: Odum School of Ecology, University of Georgia, Athens GA 30602, USA

• These authors contributed equally to this work.

Introduction

Mosquito-borne dengue virus serotypes cause approximately 50 million cases of dengue fever per year, 500,000 cases of dengue hemorrhagic fever (DHF) or dengue shock syndrome (DSS), and result in approximately 12,500 fatalities annually [1,2]. Since the 1950s, the incidence of DHF/DSS has increased over 500-fold [2], due to increases in human population, uncontrolled urbanization and international travel [3]. The major vector for dengue is the mosquito *Aedes aegypti* which thrives in households with open, water-filled containers in which larvae develop. Lack of reliable piped water service and garbage disposal systems in endemic

subtropical and tropical countries provide mosquito vectors with ample development sites [4].

Presently, there is no commercially available clinical cure for dengue and no vaccine has successfully completed clinical trials [5], leaving vector control as the only viable option for dengue prevention. Several practices are used to control dengue vector populations, including reduction or elimination of larval development sites and insecticides targeting immatures or adults. In the case of *Ae. aegypti*, the Container Inhabiting Mosquito Simulation Model (CIMSiM) [6,7] is the most detailed tool available for understanding population dynamics and the expected effects of different intervention strategies on adult female densities.

Author Summary

Dengue is a viral disease that affects approximately 50 million people annually, and is estimated to result in 12.500 fatalities. Dengue viruses are vectored by mosquitoes, predominantly by the species Aedes aegypti. Because there is currently no vaccine or specific treatment, the only available strategy to reduce dengue transmission is to control the populations of these mosquitoes. This can be achieved by traditional approaches such as insecticides, or by recently developed genetic methods that propose the release of mosquitoes genetically engineered to be unable to transmit dengue viruses. The expected outcome of different control strategies can be compared by simulating the population dynamics and genetics of mosquitoes at a given location. Development of optimal control strategies can then be guided by the modeling approach. To that end, we introduce a new modeling tool called Skeeter Buster. This model describes the dynamics and the genetics of Ae. aegypti populations at a very fine scale, simulating the contents of individual houses, and even the individual water-holding containers in which mosquito larvae reside. Skeeter Buster can be used to compare the predicted outcomes of multiple control strategies, traditional or genetic, making it an important tool in the fight against dengue.

CIMSiM is a weather-driven, dynamic life table simulation model of Ae. aegypti populations that incorporates a high level of detail about the life history of this species. Results from CIMSiM are used as the entomological input of a companion model, DENSiM [8], that models dengue transmission dynamics based on the mosquito population dynamics simulated by CIMSiM. CIMSiM and DENSiM have proven useful in characterizing local Aedes aegypti population dynamics [9] and predicting general impacts of control measures on dengue prevalence and incidence [10,11]. Despite its considerable detail, three things that CIMSiM does not take into account are spatial heterogeneity in habitat availability, potential impacts of stochastic effects - both of which could significantly affect population dynamics - and the genetics of the simulated population. A stochastic spatial model of Ae. aegypti population dynamics has also been developed separately [12,13] that does not include any genetic component.

A lack of a genetic modelling is not critical when dealing with most conventional methods of vector control unless evolution of insecticide resistance is of concern. Recent advances in molecular biology and genetic engineering, however, have presented the possibility of employing a number of control methods based on genetically engineered mosquitoes [14]. Genetic strategies fall into two broad categories: population suppression and population replacement. Population suppression methods, such as the Release of Insects carrying a Dominant Lethal (RIDL), which is a form of the sterile insect technique [15,16], aim to reduce the density of vectors by releasing genetically engineered male mosquitoes that mate with native females and cause mortality of offspring before they emerge as adults. Population replacement strategies aim to replace the resident, competent vector population with mosquitoes that are genetically engineered to not transmit a pathogen [17-19]. For both approaches a model that can predict the outcome of releasing an engineered strain in a given location and across different ecological and epidemiological circumstances is critically needed to provide guidance for which particular approach (or combination thereof) would be the most effective and to anticipate any undesirable outcome.

To address this need, we developed a modeling tool, Skeeter Buster, that can predict how Ae. aegypti population dynamics and population genetics might be affected by stochasticity and spatial variation in Ae. aegypti habitat. Skeeter Buster builds on the biologically rich components of CIMSiM, while adding stochasticity, explicit spatial structure and genetics. The construction of this model is the first step of our project that aims at evaluating conventional and genetic vector management tools and their potential success in controlling dengue incidence in human populations. To that end, Skeeter Buster will ultimately be associated with an epidemiological model. This tandem modeling tool will be comparable to the CIMSiM/DENSiM association, and will allow a direct assessment of the effects of vector control measures (including genetic approaches) on dengue prevalence and incidence. The latest version of Skeeter Buster with a userfriendly interface is available for Windows platforms at http:// www.skeeterbuster.net, and the source code is available on request from the authors.

In this first paper, we explain the characteristics and specificities of Skeeter Buster, and present results from simulations that compare the population dynamics predictions of Skeeter Buster to those of CIMSiM. Examples presented are not intended to explore the vast parameter space that is associated with this model. All parameters are, however, adjustable in the user-friendly version of the modeling tool. Subsequent articles will describe a detailed sensitivity and uncertainty analysis of this model, as well as validation against a data set of *Ae. aegypti* population dynamics in Iquitos, Peru. These detailed analyses will indicate whether there are specific details in the model that are not important for predicting the dynamics of *Ae. aegypti*, and could be dropped from the model. These analyses may also point to specific parameters in the model that have major effects on the mosquito dynamics, and therefore require better empirical estimates.

Methods

General characteristics of CIMSiM and Skeeter Buster

Because Skeeter Buster was built using many algorithms from CIMSiM, the two models share a number of identical characteristics. We schematically describe the relationships between the two models, with identical components represented in grayscale and specific additions in Skeeter Buster in color (Figure 1).

CIMSiM is a deterministic, weather-driven model that follows cohorts on a daily time scale for each immature stage (eggs, larvae, pupae) as well as female adults. Because they are not considered to impact the population dynamics, adult males are not modeled in CIMSiM. Environmental parameters include daily weather data (temperature, precipitation, and relative humidity) and an external input of food into containers. Based on these variables, CIMSiM calculates the number of individuals within all cohorts present in the model at a given time, their cumulative physiological development, weight, fecundity and gonotrophic status, as well as the transitions between life stages. With the exception of food input, all calculations in CIMSiM are applied to cohorts in a strictly deterministic fashion.

CIMSiM calculates water temperature and water level in all containers based on the available local weather data. The amount of food available in each container is also calculated daily. In addition to food depletion by consumption, three factors affect the amount of food: the external food input, a daily decay factor and the conversion of dead immatures to nutritional resources. The daily survival probability for each life stage includes a temperature-dependent component. Cumulative physiological development of each life stage is also based on temperature using an



Figure 1. Schematic representation of the general structure of CIMSiM and Skeeter Buster. C1, C2 and C3 are representative containers of three different types. E: eggs – L: larvae – P: pupae. Solid arrows represent transition of cohorts between life stages. Dashed arrows represent oviposition. Grayscale items represent model parts that are identical in CIMSiM and Skeeter Buster. Colored items are specific to Skeeter Buster. Multiple properties are modeled in Skeeter Buster. Only 4 properties are represented in this schematic figure, but the number of simulated properties can be (and typically is) much higher. Orange boxes and lines represent male adults and their biology. Red arrows represent mating, which is restricted to individuals present at the same property. Dash-dotted orange and blue lines represent male and female dispersal, respectively. Although dispersal may occur between all neighboring properties, to improve clarity it is only depicted between properties 1 and 3 for males, and between properties 2 and 4 for females. Dash-dotted green line represents container displacement. Although displacement may occur for all containers and towards any property, for clarity it is depicted only once. doi:10.1371/journal.pntd.0000508.q001

enzyme kinetics approach [20] assuming that a single enzyme determines the development rate of the insect (see equations in Text S2.2). Completion of physiological development at a given stage is attained when cumulative development reaches a threshold value (specific for each life stage). Hatch of embryonated eggs is determined by water level and water temperature in the container. Larval weight is modeled in parallel with the amount of food in each container according to the equations in [21](see in Text S2.4). Pupation requires larvae to complete physiological development as well as reach a sufficient weight. Fecundity of female adults is based on their weight, and females distribute their eggs among available containers based on the size of these containers.

These general characteristics of CIMSiM are all incorporated into Skeeter Buster, but with three major differences. First, Skeeter Buster is a stochastic model. For a given event (*e.g.* survival) applied to a specific cohort, a probability is defined for the cohort, and that same probability is applied independently to all individuals within the cohort. The number of individuals to which the event occurs is obtained by drawing a number from a binomial distribution defined by that probability and the total number of individuals in the cohort. Second, Skeeter Buster models several distinct locations (hereafter called "properties"). In the simplest setup, properties are arranged on a rectangular grid, and sets of distinct water-holding containers are assigned to individual properties (indoor or outdoor location of each container is specified). Immature cohorts are associated with a specific container within a property, and emerging adults are associated with a specific property. Finally, because Skeeter Buster also models the genetics of the population, cohorts are further distinguished by genotype.

Skeeter Buster also includes a number of components lacking in CIMSiM (see Figure 1). First, because of the genetic component of Skeeter Buster, male adults are now included in the model. Consequently, an important new component is the modeling of mating in the population. Mating is restricted to individuals present at the same property. Adults can disperse from one property to another, and containers can also be transported between properties, with the assumption that egg cohorts are carried along in the container.

In the following sections, we describe the Skeeter Buster model in more detail. We first describe the dynamics within a single property and within individual containers, and then describe the spatial structure of the model and mosquito movement among properties. We provide a complete description of the processes involved in Skeeter Buster. Some of these processes are similar to those in CIMSiM and are described in [6]. Therefore, we only describe those processes briefly in the main text, and refer the reader to supporting material (Text S2, Text S4, Dataset S1) for more details about the equations and parameters that are identical to their equivalent in CIMSiM.

Local population dynamics of immature and adult Ae. aegypti

Eggs. The number of *Ae. aegypti* eggs surviving per day is determined by the current water and air temperatures, sun exposure, and water depth of the breeding container (see Fig. 3 and 4 in [6], and Text S2.7 and S2.8).

Egg hatch is one of the most intriguing and complicated parts of the biology of Ae. aegypti [22]. We illustrate how Skeeter Buster (based on CIMSiM) determines the number of eggs hatching daily in a specific breeding site (Figure 2), which is only described briefly in [6]. Freshly laid eggs have to first develop sufficiently to finish embryonation. The rate of physiological development of eggs to embryonation depends on the average water temperature (parameters are taken from [6], see Text S2.2) and development accumulates until embryonation is finished. Embryonated eggs have to fulfill two additional requirements to hatch: the average water temperature has to be above 22°C, and the eggs have to be submerged in water. If the water is warm enough and the eggs are submerged, all newly embryonated eggs hatch immediately. If the average water temperature is below 22°C when the eggs finish embryonation, none of them hatch, and they enter a "mature" state. If the water is warm enough but the eggs are not submerged, a certain proportion (19.7%) of the eggs still hatch [23], and it is assumed that those larvae drop into the water, while the remaining eggs enter the "mature" state. 59.6% of "mature eggs" hatch every day when they are submerged in water with an average temperature above 22°C. Without submergence, none of these eggs hatch.

Eggs that hatch transform into neonate larvae in the container. Neonate larvae that hatch in the same container on the same day are separated into new larval cohorts with unique genotypes (including sex). Initial weight of neonate larvae is assumed to be 0.001 mg [21] as opposed to 0.0034 mg used in CIMSiM.

Larvae. We do not model different larval instars separately. Instead, we track larval weight, fat reserves and cumulative physiological development. In order to pupate, larvae need to meet two criteria. First, larvae need to reach complete physiological maturity, *i.e.* the cumulative physiological development has to exceed a certain threshold. We call larvae that have met this criterion "developed". Then developed larvae pupate only if they have reached sufficient weight. The weight threshold that a developed larval cohort has to exceed in order to pupate is itself dependent on the cumulative physiological development of the cohort. The weight threshold is lower for physiologically older larvae. We describe first how physiological development and larval weight are modeled, then how these two parameters define the criteria for pupation. Finally we describe survival algorithms for larval cohorts.

Larval physiological development and maturation. In CIMSiM, all larvae from a cohort become developed on the same day that their cumulative physiological development reaches 0.95 or more. However, in Skeeter Buster we introduce more realistic variability in the date of development completion by allowing some portions of the larval cohort to become developed at a lower cumulative physiological development, and other portions to reach more than the mean physiological date of maturation before they become developed. We assume that the probability of becoming developed for an individual larva is a function of its cumulative physiological development of 0.89 and all larvae becoming developed above 1.17 [24]. In between these two extremes, each larva becomes developed with a probability based



Figure 2. Algorithm for the determination of egg hatch probabilities. Process flow diagram representing the algorithm that is followed each day to determine the hatch probability of *Ae. aegypti* eggs in a cohort. doi:10.1371/journal.pntd.0000508.g002

on the cumulative physiological development of the larval cohort (see equation in Text S2.3 and Figure S3). Developed larvae from each larval cohort are moved to a newly created cohort to avoid coexistence of developed and undeveloped larvae in the same cohort. This does not significantly increase the number of larval cohorts in the model at any given time because larval physiological development completion is typically spread only over two days.

weight. calculations and density-Larval food dependence. As in CIMSiM, the dynamics of larval cohort weight and the amount of larval food in a container are governed by equations based on a laboratory study of larval development [21] (see Text S2.4). Dry weights of larvae and pupae are used here in accordance with the equations in [21]. The equations determine the change of larval weights and food levels during 4 hrs intervals, which are integrated by the Euler method for 24 hrs. The original equations in [21] were designed for and validated at 26°C. The equations are supplemented in CIMSiM and Skeeter Buster by the addition of a factor that scales for different temperatures [6]. All members of a larval cohort grow in weight uniformly, and male and female larvae grow at the same rate under identical conditions.

These equations describing the dynamics of larval weight and available food govern the density-dependent competition for food among conspecific larvae within containers. Density-dependence is considered a major component of larval dynamics in container breeding mosquitoes such as Ae. aegypti [25], and is suspected to be mostly caused by indirect competition for nutritional resources. Few studies have provided a detailed description of densitydependence for Ae. aegypti [23,26], yet the details of these dynamics are critically important for vector control purposes. The degree and pattern of density-dependence in natural populations is not well known [27]. We present a simple example of the type of density-dependence existing in Skeeter Buster, showing a major impact on two important immature traits: the survival from egg to pupa, and the weight of the surviving pupae (Figure 3). By fitting a simple model of density-dependence [26] to these results, we can distinguish how density-dependence affects survival to pupae depending on the initial density (Figure 3A): at low initial densities, undercompensatory density-dependence is observed (i.e. an increase in initial density results in an increase in the number of survivors to pupa), while overcompensatory density-dependence is observed at higher initial densities (i.e. an increase in initial density results in a decrease in the number of survivors to pupa). Note that this illustration of density-dependent processes within a single container ignores interactions between early and late instars from unsynchronized cohorts, and therefore probably underestimates the effects of density-dependence.

Larval survival. Temperature-dependent larval mortality is determined according to Fig. 3 in [6] (see also Text S2.7) based on the minimum and maximum water temperature. Larvae die with a probability of 0.95 per day if the container dries up.

The most complex cause of mortality is starvation. Figure 4 illustrates the process of determining starvation survival. The definition of starvation is that the weight of the larvae in the cohort decreases from the previous to the current day. On the first day of starvation, a prefasting lipid reserve for the larval cohort is calculated, based on Eq. 9 and 10 from [21]. During starvation, the actual lipid reserves of the larvae are reduced by the weight lost, which in turn, impact larval fasting survival. Starving larvae that still retain lipid reserves have an additional 0.05 probability of mortality compared to non-starving larvae, whereas those without reserves have an additional 0.5 probability of mortality per day. Larvae that have been starving previously but are now regaining weight increase their lipid reserves and have no additional

mortality due to starvation. Whenever the amount of actual lipid reserves of a larval cohort that has been previously starving reaches its prefasting lipid reserve, the starvation period is considered to be over.

All larvae in a cohort die if they accrue a cumulative physiological development above a maximum threshold (set at 800% of the mean development period) and still have not gained enough weight to pupate. This prevents larvae from lingering in poor nutrient conditions indefinitely. Similarly, all larvae in a cohort die if their weight drops below a specific threshold for survival (arbitrarily set to 90% of the weight at hatching, *i.e.* 0.0009 mg). Dead larvae are converted into biomass for larval food on the next day, with a 0.4 conversion factor.

Pupation. Larval pupation is modeled as in CIMSiM, where developed larvae have to reach a pupation weight threshold, which is a decreasing function of both larval cumulative physiological development and temperature. The progression of larvae in larval cohorts in terms of their cumulative physiological development as well as their weight is illustrated by their growth trajectories (see Figure S4). In CIMSiM, pupation weight thresholds are identical for males and females, and the entire larval cohort pupates together when its members reach the pupation weight threshold. In Skeeter Buster, pupation weight thresholds are lower for males compared to females, which is more realistic [28]. In addition, we created separate pupation weight thresholds at which the probabilities of pupation for members of the larval cohort are 0.25, 0.5, 0.75 and 1, respectively. The pupation weight threshold calculated in CIMSiM represents the value at which this probability equals 0.5. In CIMSiM, the pupation weight threshold is based on the cumulative physiological development of the larval cohort on the previous day. In Skeeter Buster, it is based on the cumulative physiological development calculated on the current day. Larvae entering pupation die with a probability of 0.05 due to developmental abnormalities [19] and are converted to larval food with 0.4 conversion rate. The remaining pupating larvae are assumed to successfully enter the pupal stage. In Skeeter Buster, the dry weight of members of a new pupal cohort is equivalent to the dry weight of individuals in the mature larval cohort from which those pupae originated. Therefore, if multiple larval cohorts pupate in the same container during a given day, the pupae originating from each larval cohort will be transferred into corresponding separate pupal cohorts. In contrast, CIMSiM merges pupae originating from multiple larval cohorts into a joint pupal cohort with dry weight of the new pupae calculated as the average of the dry weights of the multiple larval cohorts. Skeeter Buster hence preserves the heterogeneity in pupal weight, and ultimately adult weight, which is a trait of epidemiological importance [29].

Pupae. Because mosquito pupae do not feed, completion of this developmental stage, and emergence of adults, occurs as soon as pupae reach physiological maturation. In CIMSiM, all pupae in a cohort mature on the same day when their cumulative physiological development reaches 0.95 or more. In Skeeter Buster, we assume that the probability of maturation for an individual pupa is a function of its cumulative physiological development, as described above for larvae. No pupa matures below the cumulative physiological development of 0.89 while all pupae mature above 1.17 [24]. In between these two extremes, each pupa in the cohort becomes mature with a probability based on the current cumulative physiological development of the cohort (see Text S2.3). If any of the pupae mature, a new "mature" pupal cohort is created in order to store them separately. This does not increase the number of necessary pupal cohorts significantly because pupal physiological maturation is typically only spread over two days.



Figure 3. Density-dependence within a typical container (1-gallon bucket) in Skeeter Buster. The development of a single mature egg cohort (with N_{eggs} eggs) is tracked in a single container (10 replicated simulations). A: mortality from egg to pupa, expressed as a *k*-value: if N_{pup} is the final number of surviving pupae, $k = -\log(N_{pup}/N_{eggs})$. '+' symbols represent the outcomes of individual simulations. Solid line is the fit of the model $N_{pup} = \lambda$. N_{eggs} .exp($-\alpha N_{eggs}$) [26]. Dashed line represents the point at which the slope of the curve is exactly one (N_{eggs} *=253), marking the transition from undercompensatory (slope<1) to overcompensatory (slope>1) density-dependence (see text). B: average dry weight (in mg) of the surviving pupae.

doi:10.1371/journal.pntd.0000508.g003



Figure 4. Algorithm for larval survival probability calculations. Flow diagram depicting the algorithm that determines the daily survival probability of *Ae. aegypti* larvae that are currently experiencing or recovering from starvation. PLR = Prefasting Lipid Reserve. doi:10.1371/journal.pntd.0000508.g004

Pupal survival in Skeeter Buster depends on water temperature according to Fig. 3 in [6]. Dead pupae are converted into biomass for larval food on the next day, with a 0.4 conversion rate. All surviving pupae in a mature pupal cohort emerge on the same day. Significant mortality (17%) of pupae during eclosion is assumed as in CIMSiM. Unlike CIMSiM, Skeeter Buster accounts for pupae dying during eclosion as a source of biomass for larval food.

Adults. Pupae that successfully eclose become nulliparous female or male adults. In CIMSiM, all female adults that eclose on the same day from multiple containers form a new single female adult cohort. The adult weight associated with this cohort is equal to the average weight of all contributing pupal cohorts multiplied by a conversion factor from dry weight used in larval/pupal calculations to wet weight used in adults. Therefore, small females that eclose from a suboptimal container and large females that eclose from a very productive container on the same day are merged into a joint cohort of average-sized females, for computational simplicity. Given the importance of the size of females for traits such as fecundity [30] and blood-feeding frequency [29,31], Skeeter Buster separately treats each pupal cohort from which new adults eclose. Emerging adult females are modeled individually (so that they can later be tracked in the epidemiological model), while emerging males from each pupal cohort are transferred to a new male adult cohort. The weight of newly eclosed adults is the weight of the pupae they eclosed from, converted from dry weight to wet weight.

Because the goals of CIMSiM focus on population dynamics, and males are not generally considered to contribute to the population dynamics of *Ae. aegypti*, adult males emerging from containers are discarded in CIMSiM. In contrast, a major goal of Skeeter Buster is to assess the outcome of genetic control strategies, for which males are critical, and therefore included in the model. CIMSiM also assumes that the sex ratio of emerging pupal cohorts is always exactly 1:1. However, there is compelling evidence that in stressed conditions the sex ratio of emerging pupal cohorts is significantly skewed in the direction of males, in extreme cases leading to emergence of only males [32]. Skeeter Buster can reproduce such patterns by the assumption of a lower pupation weight threshold for males than for females. In some resourcelimited settings, the majority of male larvae can reach their pupation weight threshold and successfully pupate and eclose, while many female larvae can fail to reach their higher pupation weight thresholds and die. Such a skewed sex-ratio would affect the population dynamics and genetics of the mosquito population, which might have particularly important consequences for genetic control strategies.

Daily mortalities of male and female adult mosquitoes are likely to be extremely variable and change with local conditions. We assume nominal daily mortality probabilities of female and male adults to be 0.11 and 0.23 respectively, based on field estimates [33], as opposed to 0.09 for both sexes assumed in CIMSiM. The daily mortality probability of adult mosquitoes also depends on the minimum and maximum air temperature and on saturation deficit (see Text S2.7 and S2.8). Additionally, age-dependent mortality can be set to occur in Skeeter Buster by setting an age at which senescence starts and the maximum age that an adult male or female can reach (see e.g. [32]). Between these two ages, daily mortality probability increases linearly from the base value at the onset of senescence up to 1.0 at the maximum age.

The status of female adults in the gonotrophic cycle is defined by their cumulative physiological development and is modeled using an enzyme kinetics approach similar to the one described for immature stages. We assume that females are not limited by the availability of blood meals, that hosts are always available everywhere and are homogeneous in quality (details about blood feeding behavior will be accounted for in the future epidemiological model). Female adults are assumed to oviposit when they complete their gonotrophic cycle (i.e. when their cumulative physiological development reaches a threshold value). The first gonotrophic cycle is assumed to be considerably longer than the subsequent ones because all ovarioles of a female adult have to progress from Christopher's stage I to stage II during this cycle [34]. As with CIMSiM, Skeeter Buster accounts for this difference in cycle length by requiring that females complete 100% of cumulative physiological development during the first gonotrophic cycle, while in subsequent cycles, females only need to complete 58% of cumulative physiological development (assuming that 42% of development occurs before the cycle begins).

In CIMSiM, the number of eggs laid by female adults on a specific day is an increasing linear function of the moving average of the wet weights of the last five eclosing female adult cohorts. While this method simplifies the computation, the moving average may include the wet weight of female adult cohorts that do not complete their gonotrophic cycle on the current day or even cohorts that eclosed so long ago that the individuals they represent are already all dead. Additionally, the moving average in CIMSiM is not weighted according to the relative numbers of female adults in the respective female adult cohorts. In contrast, Skeeter Buster models female adults individually, and determines the mean fecundity of ovipositing females based on their unique wet weight. The number of eggs laid by each individual adult female in each cycle is determined randomly by drawing numbers from a normal distribution of values based on the mean fecundity, with a standard deviation of 0.375 times the mean [35]. When females are over 25 days old, their mean fecundity decreases linearly (-0.437 eggs per day over 25 [36]).

We assume that females are strictly monogamous, so that each female mates only once in its lifetime. Sperm received in this mating are then used to fertilize eggs in all subsequent gonotrophic cycles. The literature on the mating behavior of *Ae. aegypti* includes conflicting opinions on the ability and frequency of adult female *Aedes aegypti* to re-mate. While some of the sources suggest that adult females mate only once during their whole lifetime [37,38], there is evidence to show that they mate at least once every four gonotrophic cycles [39,40]. We assumed a single mating because only a small portion of adult females lives until the end of four gonotrophic cycles. However, if future studies of field populations report a significant amount of polyandry, the model could be modified to allow females to have multiple mates.

The list of male cohorts available for mating at each property is compiled daily. This only includes male adult cohorts that emerged more than 2 days ago, because adult males need 48 hrs for their external genitalia to turn 180° into the position needed for mating [22]. Male adult cohorts of the same genotype are merged in this list, so that the list of available male adult cohorts for mating only comprises one male mating cohort per genotype. Because there is evidence that male body size is positively correlated with reproductive success [41], we rank each male mating cohort according to the total wet weight of all adult male individuals of that genotype. At a specific property, the probability that a male in a mating cohort with a specific genotype is chosen for mating is based on the mating rank of that genotype, which is solely based on weight (we currently assume that genotype itself does not impact male mating ability).

In Skeeter Buster, genotypes are represented as a binary sequence of loci. We allow only two alleles at any given locus. The genotypes of the offspring resulting from the mating of a female and male of particular genotypes are determined by a random sampling of the possible gametes of the two parents.

Eggs laid by adult females of the same cohort at the same property on a single day are separated by genotype and distributed to all available containers at the property. Each container receives eggs with a probability proportional to the logarithm of the container volume. A given egg batch is distributed among containers using multinomial distributions based on these probabilities, with more eggs laid in larger containers than in smaller ones [42–44]. Eggs of the same genotype that are deposited into the same container (by all females at a given property) are summed to create a new egg cohort that is unique in terms of genotype and container location.

The exact height within the containers at which an egg is deposited is important in terms of survival and hatching. *Aedes*

aegypti distributes its eggs on the sides of the containers within a band from the water level up to a few centimeters above the water level. While CIMSiM divides the height of containers into 2 cm layers, Skeeter Buster uses a finer 2 mm resolution, which conforms better to the natural size of mosquito eggs [22]. Egg cohorts are distributed into a maximum of 20 bands (*i.e.* 4 cm) above the water level, using a multinomial distribution with equal probabilities. If the water level is less than 4 cm below the top of the container, eggs are similarly distributed into the available number of bands between the water level and the top of the container.

Spatial dynamics and movement among properties

While CIMSiM models a single representative area with a default size of 1 ha, Skeeter Buster models multiple properties independently. Each property hosts a specific set of containers both inside and outside of buildings, and the immature cohorts in these containers as well as the adults emerging from those are specifically assigned to that particular property. Properties are laid out on a rectangular grid, each cell of the grid representing a single property. The grid is not associated with explicit geographic distances, and the property is the only fundamental unit of distance. Although in this paper we consider properties to be at the scale of meters (individual houses in a dense urban setting), one property in the model can be considered to be larger units such as a block of properties or a village (and parameters can be adjusted accordingly) if needed for specific questions.

Properties located on the edges of the grid have fewer immediate neighbors than those in the interior of the grid. To deal with these edge locations we employ one of three boundary assumptions. First, periodic boundaries assume that opposite borders of the grid are connected to each other to form a toric topology. Second, solid boundaries prevent mosquitoes from migrating across the border, and force them to stay in the border property. Third, with random boundaries, mosquitoes migrating across a border are reintroduced at a random location on the edges of the grid. Properties can be identified by their coordinates (x_i, y_i) on the grid. Distance between properties (x_i, y_i) and (x_j, y_j) is defined as $|x_j-x_i|+|y_j-y_i|$ (with appropriate adjustments depending on the boundary conditions). In this paper, we only report results from model runs that use solid boundary conditions.

Adults can disperse between properties. Skeeter Buster allows for both short and long range dispersal. Short range dispersal allows adult male and female mosquitoes to move to nearest neighbor properties. We assume that this movement occurs with probability 0.3 for each mosquito on each day. We estimated this probability by simulating an empirical mark-release-recapture study in Thailand [45], and measuring the necessary daily dispersal probability to match the distribution of captured marked mosquitoes found in that study (Figure 5). In the model, for each dispersing adult, one of four directions is randomly chosen, and the adult is moved to the nearest property in that direction (von Neumann neighborhood).

Adult mosquitoes can disperse to properties at a further distance in the grid by long range dispersal. There is no clear consensus in the literature about the extent of long range dispersal of adult *Ae. aegypti* (*e.g.* how often this happens, or how far adults migrate) [45– 47]. In Skeeter Buster, each adult can disperse long distances with a daily probability; we assume a default value of 0.02. A maximum distance is also defined for long range dispersal events (default value of 20 properties, corresponding to ~200 meters in a dense urban setting). Within this range, an actual distance is chosen at random (uniformly between 1 property and the maximum distance), and the destination property is chosen randomly among



Figure 5. Dispersal of a single female adult cohort in Skeeter Buster, and virtual mark-recapture experiment. (A) Dispersal of a single female adult cohort released in a single property at day 0. Only short range dispersal is allowed (daily rate = 0.3), and survival is set at the default value (daily rate = 0.89). Solid lines represent the average number of females (20 replicated simulations) found at a given distance from the release house after 1 (circles), 2 (squares), 4 (diamonds) or 6 (triangles) days, dashed lines are 95% Cls. (B) Virtual mark-release-recapture (MRR) experiment based on this single cohort. We replicate the protocol and recapture rate described in Harrington *et al.* (2005) (Table 2, line 1) for outdoor releases in Thailand, with daily recaptures for 12 consecutive days. Dark bars are the results from the virtual MRR in Skeeter Buster (+/–SD), light bars are the results doi:10.1371/journal.pntd.0000508.q005

properties situated at this particular distance. We assume that the dispersal probabilities for both short and long range dispersal are independent of age, sex [45], parous state, mating status, size or developmental percentage.

Finally, we also allow the possibility for displacement of containers from one property to another. With some daily probability, any particular container is removed from its original property and allocated to another randomly chosen property in the grid. To account for the movement of immature cohorts associated with container displacement, all egg cohorts present in a moving container remain unaltered by this process. Larval and pupal cohorts, however, are discarded. In this paper, unless otherwise specified, the daily movement probability is assumed to equal zero.

Simulation program development

To develop the Skeeter Buster simulation program, we chose to rewrite a clone of CIMSiM in C++ as a first basis, because it provides several clear advantages for model development. From a practical point of view, CIMSiM was originally written in Visual Basic, a coding language that is tied to the PC platform and that has undergone alterations that hinder recompilation of the code on recent machines. We instead chose to use standards-compliant C++ to provide maximum flexibility, e.g. in allowing the code to be ported to and run on other computer systems, and to prevent future obsolescence of the code.

Another, more important, motivation for our strategy was to provide some means to verify our simulation code, ensuring that all procedures would work in Skeeter Buster according to the algorithms presented in the original published model [6]. The complexity of the CIMSiM (or Skeeter Buster) simulation code offers many opportunities for the occurrence of coding errors; these could be difficult to identify without an independent rewrite of the code. Rewriting CIMSiM allowed us to reveal and correct some inconsistencies between the original model and presented algorithms, as well as apparent malfunctions in the original release of CIMSiM (see Text S1, Figure S1 and Figure S2). For all the above reasons, we felt that the rewriting process of CIMSiM was a necessary step prior to working with confidence when expanding the initial model to build Skeeter Buster.

We rewrote CIMSiM in C++ (hereafter refered to as C++ CIMSiM) by exactly following the algorithms described in [6]. We tested C++ CIMSiM by systematically comparing its output to the output of the original CIMSiM program with identical parameters. Whenever the output was different, we contrasted the source code of the C++ CIMSiM to the algorithms published in [6] as well as to the source code of the original CIMSiM. We corrected several coding errors in C++ CIMSiM (see Text S1). We observed several differences between the operation of the original CIMSiM program and the algorithms described in [6]. In order to verify C++ CIMSiM, we had to deliberately include these differences and coding errors from the original source code into the C++ CIMSiM source code during this testing phase. We attributed rare remaining differences in the outputs to malfunctions of the original CIMSiM executable. We were able to mimic such malfunctions by deliberately altering specific cohorts of larvae on specific occasions in C++ CIMSiM (see Text S1). Finally, we were able to match the output of the original CIMSiM executable and the output of C++ CIMSiM (Figure 6). Because C++ CIMSiM is not affected by the malfunctions in the original CIMSiM executable and is more flexible in terms of desired output, we used C++ CIMSiM in our comparisons to Skeeter Buster.

Skeeter Buster was developed by expanding and modifying this C++ code according to the model specificities described above (see Text S4 for a detailed list of modifications). A user-friendly graphical interface was developed for PC/Windows systems, and allows the user to vary parameter values. This part of the code is more specific to the particular system, but a similar interface could be developed for other systems (or could be developed in a portable framework such as Java).

Results

In this paper we present results of Skeeter Buster simulations and compare them to output from CIMSiM (using our C++ CIMSiM version). The simulations presented here use weather data from the city of Iquitos, Peru, collected from the NCDC CDO online database [48]. Iquitos is a geographically isolated city in the Amazon basin whose *Ae. aegypti* population, larval habitat composition and dengue transmission dynamics have been studied for over 9 years [49–51]. In the simulations presented here we



Figure 6. C++ **CIMSiM as a clone of the original CIMSiM.** Number of larvae generated by the original CIMSiM (squares) and the version of C++ CIMSiM (+signs) that incorporates the small errors and malfunctions detected in the original CIMSiM, showing perfect match. Weather data were collected in Iquitos, Peru during 1978. Containers used were 1 gallon buckets [6], and both models simulated an unstructured 1-ha area. All parameters were set as in [6]. doi:10.1371/journal.pntd.0000508.g006

. www.plosntds.org

Comparison between Skeeter Buster and CIMSiM

CIMSiM and Skeeter Buster handle multiple containers of the same type in different ways. While CIMSiM models a single representative container, and multiplies the results according to the density of such containers per hectare, Skeeter Buster models each container individually. In order to compare these two approaches, we first set both CIMSiM and Skeeter Buster to have the equivalent of 100 containers of each of the above three types in an area of one hectare, with completely random mating of the mosquitoes within this area. For Skeeter Buster, this was equivalent to modeling a single "property" with a 1 ha yard in which 100 containers of each type are placed. We compare the outcome of this simulation to that of CIMSiM set up with the same three types of containers, each with a density of 100/ha. Both approaches model a similar 1-ha area. The primary difference is that Skeeter Buster models the dynamics in each of the 300 containers individually, whereas CIMSiM simulates the dynamics in groups of only 3 representative containers.

We compared the stage-specific densities of eggs, larvae, pupae and adult females within the 1-ha area from single runs of both Skeeter Buster and CIMSiM (Figure 7). For all developmental stages of Ae. aegypti, a common characteristic of the output from Skeeter Buster is that the temporal variation in density is reduced compared to CIMSiM. Although it may appear paradoxical to observe less variation in a stochastic model, this result can be explained by two major differences between these two models. First, because of the stochasticity incorporated in Skeeter Buster, the demographic dynamics in each container are independent and not synchronized, which reduces the variability when the total density across all 300 containers is considered. Second, in Skeeter Buster, individuals within a given larval cohort do not necessarily all pupate on the same day, and pupation can be spread across several days. The same effect applies for larvae maturation and pupae maturation. As a result, the 'cohort effect' is quickly lost in the simulation, reducing the temporal variation in densities.

The average stage-specific densities, taken over the entire year, in Skeeter Buster are similar to those obtained from CIMSiM.



Figure 7. Time series comparisons between C++ CIMSiM and Skeeter Buster. Stage-specific time series from C++ CIMSiM (light gray line) and Skeeter Buster (black line). Containers are 1-gallon buckets, each simulation is set up with 100 containers in a single location. Weather data used were collected in lquitos, Peru, 1978. A: Eggs; B: larvae; C: pupae; D: female adults. doi:10.1371/journal.pntd.0000508.g007

Minor differences in average densities can be explained by the different daily mortality rates used in Skeeter Buster, or by minor changes in the oviposition procedures (see Text S4). These changes also affect the periodicity of these time series, with the interval between peaks of female adult densities appearing to be slightly shorter in Skeeter Buster (see Figure 7D, and Text S3, Figure S5 and Figure S6 for a more detailed analysis of time series periodicity).

Impact of spatial structure and heterogeneity

We incorporate spatial structure in Skeeter Buster by considering simulations using the same 300 containers (100 of each type) as before, but now distributed among 100 individual properties. Properties are laid out on a 10×10 grid, and migration between individual properties can occur (see Methods). To explore the impact of habitat heterogeneity, we consider two container distributions. First, a homogeneous container distribution in which each property has exactly 3 containers, 1 of each type; in other words, all properties have an identical container distribution. Second, a heterogeneous container distribution, in which all 300 containers are randomly assigned to one of the 100 properties. In this case, the overall number of containers remains the same as in the homogenous case, but individual properties can have different types and numbers of containers.

We present a snapshot of the spatial variation in the density of the population, as the number of pupae per property, at the end of a 1-yr simulation with the homogeneous container distribution, on Figure 8. Because of the effects of both stochasticity in local dynamics and dispersal, there is clear spatial heterogeneity among population densities between individual properties, even when their container composition is the same. We compare the time



Figure 8. Spatial representation of the population simulated by Skeeter Buster. Spatial representation of the pupal composition of the population after a 1-year simulation using 100 properties each containing 3 containers (one of each type described in the text). Each square represents an individual property. The grayscale represents the number of pupae found at this property on day 365. Properties with no pupae are colored in white, whereas the presence of pupae is denoted in gray, with darker shades representing higher numbers of pupae. doi:10.1371/journal.pntd.0000508.g008

series of female adult density in the whole population for both types of habitat heterogeneity described above, as well as for the non-spatial case described in the previous section (Figure 9). Both average densities and temporal variances are comparable in all three cases, and therefore do not appear to be affected by habitat heterogeneity.

Habitat heterogeneity however has a strong effect on the level of spatial variation (between properties) in the population. We quantify this variation by measuring the coefficient of variation in the number of pupae among individual properties at a given time (denoted as CV_p). We measured CV_p (Figure 10) in the two above-defined setups (homogeneous or heterogeneous), and under three different assumptions about adult dispersal between properties : (1) both short range and long range dispersal are allowed, with daily probabilities of 0.3 and 0.02, respectively; (2) only short range dispersal is allowed, or: (3) no dispersal at all. The results of analysis of variance for CV_p are also summarized (Table 1).

These results show a clear effect of the spatial distribution of containers on CV_p . As expected, the values of CV_p are significantly higher when the container distribution is heterogeneous. Dispersal also has a significant effect. For both container distributions, CV_p is significantly higher when no dispersal occurs. On the other hand, the values of CV_p when short and long range dispersal occur do not differ from the case when only short range dispersal is allowed, suggesting that long range dispersal does not affect spatial variance among properties within the specified level of heterogeneity. Similarly, there is a significant effect of the interaction between container distribution and dispersal pattern. The effects of dispersal on CV_p are more pronounced when the container setup is heterogeneous.

Effects of adult movement and habitat heterogeneity on genetic structure of the population

Finally, we describe how the genetic structure of the population is affected by spatial factors such as the distribution of containers (homogeneous or heterogeneous) and adult dispersal. We follow the dynamics of a single locus with two alleles that do not differentially impact fitness (i.e. two neutral alleles). Both alleles are initially introduced into the population in egg cohorts homozygous for one of the two alleles, each at a frequency of 0.5. Simulations are set up with 400 properties $(20 \times 20 \text{ grid})$, with the same three container types as above, and run for 5 years. We arbitrarily define 25 subpopulations that consist of non-overlapping 4×4 squares within the 20×20 grid. (Here, we use 400 properties instead of 100 to allow us to partition our grid into a larger number of subpopulations, facilitating the spatial analysis that follows.) Short range dispersal is set to its default value (0.3 daily dispersal probability), and we examine the effects of varied amounts of long range adult and container movement on the genetic structure of the population.

We calculated the global F_{ST} values based on this neutral locus at the end of the simulations (Figure 11). F_{ST} values, representing the level of genetic differentiation within the overall population (between subpopulations), are higher in the case of a heterogeneous distribution of containers, but decrease quickly when the daily probability of long range dispersal increases.

We also calculate pairwise F_{ST} values between all 25 subpopulations. We can test the existence of isolation by distance in our simulated population by examining the correlation between the genetic distance between two subpopulations (given by the pairwise F_{ST} value) and their geographic distance. More specifically, following the method described in [52], we regress the values of $F_{ST}/(1-F_{ST})$ for pairs of subpopulations against the logarithms of their geographic distances. Isolation by distance is



Figure 9. Effects of habitat heterogeneity on female adult densities. Time series of the total number of female adults in the population from three different Skeeter Buster simulations. All three setups use 100 containers of each type (see text for description). Non-spatial (gray line) is a single property containing all 300 containers. Homogeneous distribution (blue line) means 100 properties each containing exactly 3 containers (one of each type). Heterogeneous distribution (red line) means 100 properties with the 300 containers randomly distributed amongst them. In all cases, the model is initialized with egg cohorts only. The results presented are for year 2 of the simulation. Respective mean number of females +/–SD are: homogeneous: 413.8+/–46.5 ; heterogeneous: 401.9+/–47.3 ; 425.4+/–57.4. doi:10.1371/journal.pntd.0000508.q009

characterized by a significant correlation between these two distances. A stronger isolation by distance is associated with a higher slope of the regression line. We measured the values of this slope for different assumptions concerning habitat heterogeneity and adult movement (Figure 12). For both types of container distribution, long range dispersal, even at daily probabilities as low as 0.02, prevents the occurrence of isolation by distance at the scale of the simulation considered here (20×20 properties).

Finally, we also examine the impact of container displacement (and the associated movement of egg cohorts) between properties. We measured the impact of this movement on final F_{ST} values for a neutral allele, assuming that there is no long range dispersal (Figure 13). Only the plastic buckets (1-gallon and 5-gallon) are moved since larger containers are not typically moved among households. It appears that moving containers across the city can have an impact on the population structure even when these events are rare, although increasing this probability does not seem to impact F_{ST} values as much as adult dispersal.

Discussion

The results from Skeeter Buster presented in this paper using simplified container and property setups highlight the impact of spatial structure and heterogeneity on the population dynamics of *Ae. aegypti*. First, the simulated population dynamics differ markedly between CIMSiM and Skeeter Buster when a large number of identical containers within one property are considered in Skeeter Buster. Because each of these containers is simulated individually in the stochastic Skeeter Buster, the overall population dynamics is an average over a large number of containers whose individual dynamics are typically not synchronized. Additionally, containers in different properties are associated with a different local population. Identical containers in Skeeter Buster can therefore exhibit very different dynamics from one another. As a consequence, the variability in densities of *Ae. aegypti* at the level of the population is greatly reduced (see Fig. 7).

Beyond the effect of simulating individual containers, the explicit simulation of individual properties in Skeeter Buster does



Figure 10. Effects of habitat heterogeneity and adult dispersal on density heterogeneity between properties. Effects of habitat heterogeneity and adult dispersal on the spatial coefficient of variation of the number of pupae among properties (CV_p). Two container distributions are considered (see text for details): homogeneous (blue line and boxes), or heterogeneous (red line and boxes). Three adult dispersal patterns are considered: (i) no dispersal; (ii) short range dispersal (SRD) only, and (iii) both short range dispersal and long range dispersal (LRD). For both (ii) and (iii), short range dispersal occurs with daily probability of 0.3 per adult per day. For (iii), long range dispersal occurs with daily probability of 0.02 per adult per day. For each combination (container distribution×dispersal pattern), 50 simulations are run. For each simulation, the plotted value of CV_p is calculated as the average of the daily CV_p value for the last 100 days of the simulation. The result from each simulation is represented by a 'x' symbol. Boxes show, for each combination, the 25% and 75% quantiles. The middle line in the box represents the median and the whiskers encompass the data points that fall within 1.5 interquartile range in each direction. The lines between boxes connect the means. Within each container setup, pairwise mean comparisons are tested by Student's t-test (NS: p>0.05; ****: p<0.001).

not seem to affect the overall population dynamics, at least in the settings investigated here (see Fig. 9). However, this inclusion of multiple properties allows a quantitative description of spatial heterogeneity among properties in terms of *Ae. aegypti* densities and age composition that could not be modeled by CIMSiM. We show here that the level of heterogeneity among properties in *Ae. aegypti* population density can be high even when a homogeneous container distribution is considered. Future studies based on Skeeter Buster will reveal if and how much this heterogeneity is predicted to affect both dengue transmission dynamics and the impact of vector control strategies. Because there is evidence that heterogeneity among properties in densities of female adults could be important for both [49,53], we conclude that it is an important feature to include in our modeling tool.

Among the possible strategies for decreasing dengue incidence, approaches using genetic tools to control the mosquito population appear to be promising, but their applicability in field situations is still under evaluation. Skeeter Buster was designed to aid this evaluation, and simulate the efficiency and practicality of these approaches in order to guide the development of genetic control programs. We therefore incorporated explicit genetics in the model, and describe here the basic population genetic structure predicted by this model. While long range dispersal does not seem to affect the spatial variance in densities, Figs. 11 and 12 show that long range dispersal can significantly affect the genetic structure of the population. Even relatively rare long range dispersal events (daily probability lower than 2%) are associated with lower F_{ST} values in the population and dramatically reduce the observed

. www.plosntds.org

Table 1. Analysis of variance in CV_p values testing for the effects of habitat heterogeneity dispersal pattern.

Source	df	Sum of squares	F	р
Habitat heterogeneity	1	7.413	41713.1	< 0.0001
Dispersal	2	0.446	1254.16	< 0.0001
Habitat het.×Dispersal	2	0.210	591.31	< 0.0001
Error	294	0.052		
Total	299	8.121		

 CV_p = coefficient of variation in the number of pupae among individual properties. Habitat heterogeneity can be homogeneous or heterogeneous. Types of dispersal can be: no dispersal, short range dispersal only, or both short and long range dispersal. df = degrees of freedom, F = F-statistic. doi:10.1371/journal.pntd.0000508.t001

instances of isolation by distance among subdivisions of our modeled population. The transfer of containers between properties in the grid can also impact the genetic structure, although its impact does not seem to be as important as that of adult long range dispersal (Figure 13). The existence of strong genetic spatial structure in the population is important to the potential fate of an allele introduced into specific locations within a population. Strong genetic structure could impede or slow the spread of a novel allele to distant parts of the population. For this reason, the ability of Skeeter Buster to model this genetic structure is an important addition for predicting the outcome of genetic control strategies in *Ae. aegypti* populations.

The spatial scale examined in Skeeter Buster is at the level of individual properties, that is, in the case of Iquitos, distances of an order of magnitude of hundreds of meters. Field studies of genetic structure at this level are rare. $F_{\rm ST}$ values reported from small-scale clusters (kilometers) in within-city studies [54–57] are variable but

consistent with the highest values observed in the simulations presented in this paper. This would suggest a limited amount of adult dispersal between these geographically close sites, without excluding the possibility of gene flow maintained by displacement of immatures through human activities and transportation. More generally, these results emphasize the need to characterize the dispersal patterns of *Ae. aegypti* in natural populations. While adults are generally considered to migrate only short distances (modeled by our short range dispersal) [45], dispersal to longer distances has been observed [46], but how often such long range dispersal events occur is unknown.

Overall, the results presented here are consistent with our assertion that Skeeter Buster provides a realistic description of Ae. *aegypti* population dynamics and will be a valuable tool in the development of city-wide genetic strategies for prevention of dengue and control of its major mosquito vector. Ultimately, this entomological simulation will be a component of a framework from which dengue transmission can be modeled, and control measures can be evaluated. However, two important requirements have to be fulfilled before these further steps can be carried out. First, the outcome of the model must be validated with population data from an actual field site: this will rely on a more elaborate property setup and container distribution than the examples presented here. Skeeter Buster allows for detail at the individual container level, and therefore enables a specific Ae. aegypti population in a particular location to be modeled. But, to achieve such a location-specific level of accuracy, Skeeter Buster requires intensive field work to obtain a description of the container distribution and relative productivity in this particular location. In a subsequent paper, we will illustrate this location-specific simulation capacity with a case study of the city of Iquitos, Peru. Second, since this model relies on a very high number of procedures and parameters, all of which are associated with some level of uncertainty, it is crucial to carry a broad-scale uncertainty



Figure 11. Effects of adult dispersal on the genetic structure of the population. Final F_{ST} values after a 5-year simulation of a 20×20 grid of properties subdivided in 4×4 squares. Calculation is based on one neutral marker, and two alleles introduced at equal frequency at every property in the population. Solid diamonds: homogeneous container distribution; Open circles: heterogeneous container distribution (mean+/-SD). LRD = long range dispersal. 20 simulations are run for each container distribution×dispersal assumption combination. doi:10.1371/journal.pntd.0000508.g011



Figure 12. Effects of adult dispersal on the existence of isolation by distance in the population. Slope of the linear regression of $F_{ST}/(1-F_{ST})$ against the geographic distance between pair of subpopulations at the end of the simulation (see Fig. 10 for a description of the simulations), as a function of the daily probability of long range dispersal. Solid diamonds: homogeneous container distribution ; Open circles: heterogeneous container distribution.

doi:10.1371/journal.pntd.0000508.g012



Figure 13. Effects of container movement probability on the genetic structure of the population. Final F_{ST} values after a 5-year run (see Fig. 10 for a description of the simulation). Symbols are the average value across 10 repetitions, error bars are SD. Container movement probability is the daily probability for each container of being moved to another property. Because container movement would rapidly lead an initially homogeneous container distribution to become heterogeneous, here we only present simulations that employ the heterogeneous container distribution.

doi:10.1371/journal.pntd.0000508.g013

and sensitivity analysis of this model. This analysis will ensure that the results of the simulations are robust enough within the range of the existing uncertainties on parameter values, or, if not, the analysis will highlight the traits predicted to account for the highest percentage of uncertainty in predicted population dynamics and genetics, providing guidelines for the most needed additional field or laboratory studies.

Supporting Information

Dataset S1 Spreadsheet that references and summarizes relevant publications on *Ae. aegypti*.

Found at: doi:10.1371/journal.pntd.0000508.s001 (0.22 MB XLS)

Figure S1 Discrepancies between uncorrected C++ CIMSiM and original CIMSiM. Number of larvae for C++ CIMSiM (green) without the corrections detailed in the text and in the absence of manipulations, and for the original CIMSiM (red). Weather data was collected for Iquitos, Peru 1978. Containers used were 1 gallon buckets.

Found at: doi:10.1371/journal.pntd.0000508.s002 (0.03 MB TIF)

Figure S2 Details of the discrepancies between uncorrected C++ CIMSiM and original CIMSiM and associated cohort manipulations. Differences in the number of eggs (red), larvae (green) and pupae (blue) between uncorrected C++ CIMSiM (in the absence of the cohort manipulations discussed in the text) and the original CIMSiM. Weather data was collected for Iquitos, Peru 1978. Containers used were 1 gallon buckets. Arrows mark days 158 and 232 at which malfunctions occur in the original CIMSiM.

Found at: doi:10.1371/journal.pntd.0000508.s003 (0.08 MB TIF)

Figure S3 Cumulative proportion of larvae reaching physiological development based on the current physiological status of the cohort. For values of CDt (cumulative physiological development) between 0.89 and 1.17, a certain proportion of larvae within the cohort can become developed. In Skeeter Buster, the actual number of larvae becoming developed is drawn from a binomial distribution (see the calculation of the probability associated to this distribution in the text). Note that 50% of larvae are expected to become mature before the cumulative physiological development reaches 1.0, and 50% after.

Found at: doi:10.1371/journal.pntd.0000508.s004 (0.03 MB TIF)

Figure S4 Pupation windows as a function of physiological development status and temperature. Pupation windows define the required minimal larval weight for pupation. Lines correspond, from top to bottom, to temperatures of 15, 20, 25, 30 and 35°C. Symbols represent hypothetical larval trajectories from simulation using weather data from Iquitos, Peru, under different nutritional conditions (dark blue: high food; red: medium; light blue: low food) and show the progress of these cohorts in terms of weight gain and cumulative physiological development (CDt). Pupation of these larvae occurs when their trajectory crosses the pupation window. Found at: doi:10.1371/journal.pntd.0000508.s005 (0.01 MB EPS)

Figure S5 Periodogram of female adult density from C++ CIMSiM and Skeeter Buster. This periodogram is based on a discrete Fourier transformation of the time series presented in the

References

- WHO (2009) Dengue and dengue haemorrhagic fever Fact Sheet 117. http:// www.who.int/mediacentre/factsheets/fs117/en/.
- Kyle JL, Harris E (2008) Global Spread and Persistence of Dengue. Ann Rev Microbiol 62: 71–92.
- Gubler DJ (1997) Dengue and dengue hemorrhagic fever: its history and resurgence as a global public health problem. In: Gubler DJ, Kuno G, eds (1997) Dengue and dengue hemorrhagic fever. London, UK: CAB International.

main text (Fig. 6D). The dominant period of the cycles is approximately two days shorter in Skeeter Buster, resulting in approximately 13 density peaks a year, compared to the 12 peaks predicted by C++ CIMSiM.

Found at: doi:10.1371/journal.pntd.0000508.s006 (0.02 MB TIF)

Figure S6 Periodograms of female adult densities for various setups of Skeeter Buster. These periodograms are based on discrete Fourier transformation of time series from the model. All simulations are run using 1-gallon buckets and weather data from Iquitos, Peru, 1978-1980. Simulations are run for three years. To avoid initial cohort effects, only the last two years of each time series is analyzed. Moreover, the population is initialized with cohorts from all life stages, in proportions defined by a run of Skeeter Buster with non-limiting food. The left column represents simulations with no spatial structure, and 100 containers within the same location. The right column represents simulations with spatial structure, and 100 properties, each containing one single container. In the latter case, only short range dispersal is allowed (there is no long range dispersal). Rows correspond to different food conditions, modeled as daily food gain per container: top row, low food amounts (0.8 mg/day) ; middle row, medium (default) food amount (1.8 mg/day) ; bottom row, high food amount (3.0 mg/day). Note that the y-axes have different scales between panels.

Found at: doi:10.1371/journal.pntd.0000508.s007 (0.07 MB TIF)

Text S1 Verification of C++ CIMSiM against the original CIMSiM.

Found at: doi:10.1371/journal.pntd.0000508.s008 (0.03 MB DOC)

Text S2 Details of CIMSiM elements used in Skeeter Buster, together with modifications adopted.

Found at: doi:10.1371/journal.pntd.0000508.s009 (0.13 MB DOC)

Text S3 Analyses of periodicities in the time series.

Found at: doi:10.1371/journal.pntd.0000508.s010 (0.02 MB DOC)

Text S4 List of differences in biological procedures between CIMSiM and Skeeter Buster.

Found at: doi:10.1371/journal.pntd.0000508.s011 (0.05 MB DOC)

Acknowledgments

We are grateful to A. Garcia, Y. Huang and C. Ward for their help with the development of the model, and to A. Ellis as well as two anonymous reviewers for valuable comments on the manuscript. Any opinions expressed here reflect solely the views of the authors, and do not necessarily represent the official views of NIAID, NIH.

Author Contributions

Conceived and designed the experiments: KM TWS ALL FG. Analyzed the data: ML ALL FG. Contributed reagents/materials/analysis tools: DAF. Wrote the paper: KM ML. Built the model: KM ML MEP ALL. Revised the paper: MEP TWS ALL FG.

- Knudsen AB, Slooff R (1992) Vector-borne disease problems in rapid urbanization - New approaches to vector control. Bull WHO 70: 1–6.
- Monath TP (2007) Dengue and yellow fever-challenges for the development and use of vaccines. New Engl J Med 357: 2222–2225.
- Focks DA, Haile DG, Daniels E, Mount GA (1993) Dynamic life table model of <u>Aedes aegypti</u> (Diptera: Culicidae) - Analysis of the literature and model development. J Med Entomol 30: 1003–1017.

- Focks DA, Haile DG, Daniels E, Mount GA (1993) Dynamic life table model for *Aedes aegypti* (Diptera: Culicidae) - Simulation and validation. J Med Entomol 30: 1018–1028.
- Focks DA, Daniels E, Haile DG, Keesling JE (1995) A simulation model of the epidemiology of urban dengue fever - Literature analysis, model development, preliminary validation and samples of simulation results. Am J Trop Med Hyg 53: 489–506.
- Williams CR, Johnson PH, Long SA, Rapley LP, Ritchie SA (2008) Rapid estimation of *Aedes aegypti* population size using simulation modeling, with a novel approach to calibration and field validation. J Med Entomol 45: 1173–1179.
- Focks DA, Brenner RJ, Hayes J, Daniels E (2000) Transmission thresholds for dengue in terms of *Aedes aegypti* pupae per person with discussion of their utility in source reduction efforts. Am J Trop Med Hyg 62: 11–18.
- Morrison AC, Zielinski-Gutierrez E, Scott TW, Rosenberg R (2008) Defining challenges and proposing solutions for control of the virus vector *Aedes aegypti*. PLoS Med 5: 362–366.
- Otero M, Solari HG, Schweigmann N (2006) A stochastic population dynamics model for Acdes aegypti: Formulation and application to a city with temperate climate. Bull Math Biol 68: 1945–1974.
- Otero M, Schweigmann N, Solari HG (2008) A stochastic spatial dynamical model for Aedes aegypti. Bull Math Biol 70: 1297–1325.
- Gould F, Magori K, Huang YX (2006) Genetic strategies for controlling mosquito-borne diseases. Am Sci 94: 238–246.
- Thomas DD, Donnelly CA, Wood RJ, Alphey LS (2000) Insect population control using a dominant, repressible, lethal genetic system. Science 287: 2474–2476.
- Phuc HK, Andreasen MH, Burton RS, Vass C, Epton MJ, et al. (2007) Lateacting dominant lethal genetic systems and mosquito control. BMC Biol 5: 11.
- James AA (2005) Gene drive systems in mosquitoes: rules of the road. Trends Parasitol 21: 64–67.
- Franz AWE, Sanchez-Vargas I, Adelman ZN, Blair CD, Beaty BJ, et al. (2006) Engineering RNA interference-based resistance to dengue virus type 2 in genetically modified *Aedes aegypti*. Proc Natl Acad Sci U S A 103: 4198–4203.
- Sinkins SP, Gould F (2006) Gene drive systems for insect disease vectors. Nat Rev Genet 7: 427–435.
- Sharpe PJH, DeMichele DW (1977) Reaction kinetics of poikilotherm development. J Theor Biol 64: 649–670.
- Gilpin ME, McClelland GAH (1979) Systems-analysis of the yellow fever mosquito Aedes aegypti. Forts Zool 25: 355–388.
- Christophers SR (1960) Aedes aegypti (L.), the yellow fever mosquito. Cambridge, UK: Cambridge University Press.
- Southwood TRE, Tonn RJ, Yasuno M, Reader PM, Murdie G (1972) Studies on life budget of *Aedes aegypti* in Wat Samphaya, Bangkok, Thailand. Bull WHO 46: 211–226.
- Rueda LM, Patel KJ, Axtell RC, Stinner RE (1990) Temperature-dependent development and survival rates of *Culex quinquefasciatus* and *Aedes aegypti* (Diptera: Culicidae). J Med Entomol 27: 892–898.
- Service MW (1985) Population dynamics and mortalities of mosquito preadults. In: Lounibos LP, Rey JR, Frank JH, eds (1985) Ecology of mosquitoes: Proceedings of a workshop. Vero Beach, FL: Florida Medical Entomology Laboratory.
- Dye C (1984) Models for the population dynamics of the yellow fever mosquito, *Aedes aegypti*. J Anim Ecol 53: 247–268.
- Legros M, Lloyd AL, Huang YX, Gould F (2009) Density-dependent intraspecific competition in the larval stage of *Aedes aegypti* (Diptera: Culicidae): revisiting the current paradigm. J Med Entomol 46: 409–419.
- Barbosa P, Peters TM, Greenough NC (1972) Overcrowding of mosquito populations: Responses of larval *Aedes aegypti* to stress. Environ Entomol 1: 89–93.
- Nasci RS (1986) The size of emerging and host-seeking *Aedes aegypti* and the relation of size to blood-feeding success in the field. J Am Mosquito Contr 2: 61–62.
- Briegel H (1990) Metabolic relationship between female body size, reserves and fecundity of *Aedes aegypti*. J Insect Physiol 36: 165–172.
- Scott TW, Amerasinghe PH, Morrison AC, Lorenz LH, Clark GG, et al. (2000) Longitudinal studies of Aedes aegypti (Diptera : Culicidae) in Thailand and Puerto Rico: Blood feeding frequency. J Med Entomol 37: 89–101.
- Wada Y (1965) Effect of larval density on the development of Aedes aegypti (L.) and the size of adults. Quaestiones entomologicae 1: 223–249.

- Simulation Model of Ae. aegypti Populations
- McDonald PT (1977) Population characteristics of domestic Aedes aegypti (Diptera: Culicidae) in villages on Kenya coast. 1. Adult survival and population size. J Med Entomol 14: 42–48.
- Macdonald WW (1956) Aedes aegypti in Malaya. II. Larval and adult biology. Ann Trop Med Parasitol 50: 399–414.
- Naksathit AT, Scott TW (1998) Effects of female size on fecundity and survivorship of *Aedes aegypti* fed only human blood versus human blood plus sugar. J Am Mosquito Contr 14: 148–152.
- Styer LM, Carey JR, Wang JL, Scott TW (2007) Mosquitoes do senesce: Departure from the paradigm of constant mortality. Am J Trop Med Hyg 76: 111–117.
- Craig GB (1967) Mosquitoes: female monogamy induced by male accessory gland substance. Science 156: 1499–1501.
- George JA (1967) Effect of mating sequence on egg-hatch from female Aedes aegypti (L.) mated with irradiated and normal males. Mosq News 27: 82–86.
- Williams RW, Berger A (1980) The relation of female polygamy to gonotrophic activity in the Rock strain of *Aedes aegypti*. Mosq News 40: 597–604.
- Young ADM, Downe AER (1982) Renewal of sexual receptivity in mated female mosquitoes *Aedes aegypti*. Physiol Entomol 7: 467–471.
- Ponlanwat A, Harrington LC (2007) Age and body size influence male sperm capacity of the dengue vector *Aedes aegypti* (Diptera: Culicidae). J Med Entomol 44: 422–426.
- Surtees G (1960) Studies in the oviposition behavior and development of the type form of *Aedes aegypti* (Diptera: Culicidae).
 Breeding site dimensions. Entomol Mon Mag 95: 220–221.
- Surtees G (1967) Factors affecting oviposition of *Aedes aegypti*. Bull WHO 36: 594–596.
- Harrington LC, Ponlawat A, Edman JD, Scott TW, Vermeylen F (2008) Influence of container size, location, and time of day on oviposition patterns of the dengue vector, *Aedes aegypti*, in Thailand. Vector-Borne Zoonot 8: 415–423.
- Harrington LC, Scott TW, Lerdthusnee K, Coleman RC, Costero A, et al. (2005) Dispersal of the dengue vector *Aedes aegypti* within and between rural communities. Am J Trop Med Hyg 72: 209–220.
- Reiter P, Amador MA, Anderson RA, Clark GG (1995) Dispersal of *Aedes aegypti* in an urban area after blood-feeding as demonstrated by rubidium-marked eggs. Am J Trop Med Hyg 52: 177–179.
- 47. Trpis M, Hausermann W, Craig GB (1995) Estimates of population size, dispersal and longevity of domestic *Aedes aegypti* (Diptera: Culicidae) by markrelease-recapture in the village of Shauri Moyo in Eastern Kenya. J Med Entomol 32: 27–33.
- NCDC National Climatic Data Center Climate Data Online. http://www7. ncdc.noaa.gov/CDO/.
- Getis A, Morrison AC, Gray K, Scott TW (2003) Characteristics of the spatial pattern of the dengue vector, *Aedes aegypti*, in Iquitos, Peru. Am J Trop Med Hyg 69: 494–505.
- Morrison AC, Gray K, Getis A, Astete H, Sihuincha M, et al. (2004) Temporal and geographic patterns of *Aedes aegypti* (Diptera : Culicidae) production in Iquitos, Peru. J Med Entomol 41: 1123–1142.
- Morrison AC, Sihuincha M, Stancil JD, Zamora H, Astete H, et al. (2006) Aedes aegypti (Diptera: Culicidae) production from non-residential sites in the Amazonian city of Iquitos, Peru. Ann Trop Med Parasitol 100: S73–S86.
- 52. Rousset F (1997) Genetic differentiation and estimation of gene flow from Fstatistics under isolation by distance. Genetics 145: 1219–1228.
- Favier C, Schmit D, Muller-Graf CDM, Cazelles B, Degallier N, et al. (2005) Influence of spatial heterogeneity on an emerging infectious disease: the case of dengue epidemics. Proc R Soc Lond B Biol Sci 272: 1171–1177.
- Tran Khanh T, Vazeille-Falcoz M, Mousson L, TH H, Rodhain F, et al. (1999) *Aedes aegypti* in Ho Chi Minh City (Vietnam): susceptibility to dengue 2 virus and genetic differentiation. Trans R Soc Trop Med 93: 581–586.
- Huber K, Le Loan L, Hoang TH, Ravel S, Rodhain F, et al. (2002) Genetic differentiation of the dengue vector, *Aedes aegypti* (Ho Chi Minh City, Vietnam) using microsatellite markers. Mol Ecol 11: 1629–1635.
- Paupy C, Orsoni A, Mousson L, Huber K (2004) Comparisons of amplified fragment length polymorphism (AFLP), microsatellite, and isoenzyme markers: Population genetics of *Aedes aegypti* (Diptera : Culicidae) from Phnom Penh (Cambodia). J Med Entomol 41: 664–671.
- da Costa-Ribeiro MCV, Lourenco-De-Oliveira R, Failloux AB (2006) Higher genetic variation estimated by microsatellites compared to isoenzymes markers in *Aedes aegypti* from Rio de Janeiro. Mem I Oswaldo Cruz 101: 917–921.